Author’s response to reviews

Title: Risk of death or hospital admission among community-dwelling older adults living with dementia in Australia

Authors:

Emily (Chuanmei) You (emilyyou0127@gmail.com)
David Dunt (d.dunt@unimelb.edu.au)
Vanessa White (v.white@latrobe.edu.au)
Stephen Vander Hoorn (vander@unimelb.edu.au)
Colleen Doyle (c.doyle@nari.unimelb.edu.au)

Version: 2 Date: 29 January 2014

Author’s response to reviews:

1. Information on ethics "All fieldwork methods were approved by La Trobe University Human Ethics Committee (07-084)" has been added to the first paragraph of the Methods section.

2. An abstract has been provided following the title page.